Author’s response to reviews

Title: Stable socioeconomic inequalities in ischaemic heart disease mortality during the economic crisis: A time trend analysis in 2 Spanish settings

Authors:

Xavier Bartoll (xbartoll@aspb.cat)
Mercè Gotsens (mgotsens@aspb.cat)
Marc Mari-Dell’Olmo (mmari@aspb.cat)
Laia Palència (lpalenci@aspb.cat)
Montse Calvo (mcalvo@ej-gv.es)
Santiago Esnaola (sesnaola@gmail.com)
Carme Borrell (cborrell@aspb.cat)

Version: 2 Date: 18 Feb 2019

Author’s response to reviews:

Thank you very much for the evaluation. We have added the units of rate in the title as "(per 100.000 population)" in Tables 1 and 2.